Author's response to reviews

Title: Anti Hyperglycemic and Renal Protective Properties of Terminalia chebula Seed Powder Chloroform Extract in Streptozotocin Induced Diabetic Rats

Authors:

Nalamolu Koteswara Rao (nalamolu_k@rediffmail.com)
Srinivas Nammi (nammi@rediffmail.com)

Version: 2 Date: 8 September 2005

Author's response to reviews:

Dear Sir,

We have made the necessary corrections as per the comments given by the reviewers. please do the needful.